Author's response to reviews

Title: In vitro effect of bergamot (Citrus bergamia) juice against cagA-positive and-negative clinical isolates of Helicobacter pylori.

Authors:

   Angela Filocamo (afilocamo@unime.it)
   Carlo Bisignano (cbisignano@unime.it)
   Nadia Ferlazzo (nferlazzo@unime.it)
   Santa Cirmi (scirmi@unime.it)
   Giuseppina Mandalari (gmandalari@unime.it)
   Michele Navarra (mnavarra@unime.it)

Version: 5 Date: 4 July 2015

Author's response to reviews:

Dear Editor,

we have included a sentence in the Methods section of the revised manuscript reporting that the plant material used in our study is deposited in a publicly available herbarium.

Best wishes,

Giuseppina Mandalari